Author's response to reviews

Title: Risk of Infection and adverse outcomes among pregnant working women in selected occupational groups: A study in the Danish National Birth Cohort

Authors:

Maria Morales-Suárez-Varela (maria.m.morales@uv.es)
Linda Kaerlev (l.kaerlev@tdcadsl.dk)
Jin Liang Zhu (zjl@soci.au.dk)
Agustín Llopis-González (agustin.llopis@uv.es)
Natalia Gimeno-Clemente (natallaclemente82@hotmail.com)
Ellen A. Nohr (ean@soci.au.dk)
Jens P. Bonde (jpbb@bbh.regionh.dk)
Jorn Olsen (jo@ucla.edu)

Version: 5 Date: 12 November 2010

Author's response to reviews: see over
Dear Editorial Team,

Enclosed you will find a new version of the manuscript “Risk of Infection and adverse outcomes among pregnant working women in selected occupational groups: A study in the Danish National Birth Cohort”. In this new version we have include the four tables in the manuscript, in a tables section following the references.

Best regards,

María Morales-Suárez-Varela.